Joint Field Office Situation Report 

Human Services

Date: 9/21/05

	Registrations
	882,895

	Total Payments
	1,087,813,550.00

	Designated Parishes 
	31


	Section Reporting:  
	VOLAG

	Group Leader Contact Info:
	JON WALLACE  VAL DEPUTY I.C.

	1.
	SITUATION:

	
	A.
	General Status of Operations
	RED CROSS:

General Status of Operations:             

154,236 meals this period 4,096,110 meals to date                      

 89 shelters this period; 256 shelters to date

16,736 population period; 56,835

218,736 clients in transition accommodations and 65,224 rooms being utilized.

	
	B.
	Major Priorities Accomplished
	UNITED SIKHS :

· Helped arrange clothing and supply donations at Dream Center, Donaldsonville.  

· Delivered food supplies for distribution to families in the Greater New Orleans area. 

TSA:
 Visited Salvation Army sites in Laplace and Gonzalves today.

Major Priorities Accomplished:

Material Support Services is assessing requirements, identifying new resources to augment Red Cross stocks, purchasing material, food, and trucks in anticipation of Hurricane Rita landfall without taking resources from the Katrina effort.

Adventist Community Services – Disaster Service:

In the last 4 days

Sept-17

 Off loaded  2 trucks, top items

            New Pots

            Paper products

            Bandages

            

Sent out  2 orders - 10 pallets           

Sept-18

 Off loaded   3 trucks, top item

            water         

Sent out  1 orders - 1 pallets

Sept-19

 Off loaded  5 trucks, top items

            Baby strollers

            Cribs

            New clothing            

Sent out  30 pallets 

            Food stuffs/snacks

            Blankets

Sept-20

 Off loaded  5 trucks, top items

            Hygiene kits

            New clothing

            Toiletries

Sent out   orders - 50 pallets,  top items

            Food stuff

            Blankets

            Cots

CATHOLIC CHARITIES:

Catholic Charities yesterday secured a 60-day contract with Louisiana Department of Social Services to provide 40-80 caseworkers to help with displaced persons being placed in temporary housing. 

AMERICORPS:
8 Members of the St. Louis Safety Corps. arrived this morning to support the donations and volunteer management staff.
10 members of the National Civilian Community Corps. (NCCC) Are arriving this afternoon. 2 will be assigned to work in the JFO with the VALS; the others will be assigned where needed.

A2H:

A2H received food and other grocery products to date: 10,159,598 lbs. (310) truckloads.

LA Food Banks distributed 7,820,304 lbs. to date.

	
	C.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	8
	7 are being worked on
	8


	Section Reporting: 
	SBA

	Group Leader Contact Info:
	Carl Gaspari

	1.
	SITUATION:

	
	A.
	General Status of Operations
	Continue to hire and train customer service representatives for field deployment to existing and future DRCs.
SBA senior officials met again today with the Secretary of Louisiana Economic Development.
SBA PIO’s spoke today to a group of physicians and staffers of Memorial Hospital in New Orleans to provide SBA program information and answer questions about how a disaster loan can help them recover.

	
	B.
	Major Priorities Accomplished
	All open DRC’s are staffed with SBA representatives.

	
	C.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	10 staff at JFO

93 out in field
	0
	0

	2.
	CURRENT ISSUES OR PROBLEMS:  None


	Section Reporting: 
	HS Programs, Other Programs/Special Needs Group 

	Group Leader and Contact Info:
	Margaret Braun, Desk 225-346-4157; Cell 703-547-7768 

	1.
	SITUATION:

	
	A.
	General Status of Operations
	· Special Needs Teams (SNT) Referrals-Cumulative: 156; Pending: 7. 

· (Referral from ERO) Began coordinating resources with VOLAG for food, clothing, medicine needs of two shelters with total of 70 evacuees, of which 27are mentally challenged. Will also request CR complete site visit for registering and overall assessment; SNT will follow-up on CR report.                                                                                                                                                                    
· CC: Each DRC will be staffed with at least one CC; staff schedule being developed. Requested a copy of CC staff schedule for each DRC.

· DUA: The State is preparing a supplemental funding request as the current $11,150,000.00 obligation has been exhausted;
FEMA has obligated $25 million to expedite this process.  

· DLS: Young Lawyers Division (YLD) now has presence in several DRCs, details forthcoming. The State Bar president made an inquiry regarding the availability of phone lines in the DRC-referred to DRC Manager for response. 

· Other Federal Agencies (OFA) Liaison: Position redirected to another Group thus OFA Liaison will report activity henceforth.

· ONA: Continue to gather information and resources that may assist with more expeditious processing of funeral cases. Contacted LA Coroner’s Association President with inquiry regarding information on “confirmed” disaster related deaths; President will refer inquiry to Parish Coroners. First draft for Lump Sum award proposal completed by COB yesterday but, revisions began today.  

	
	B.
	Major Priorities Accomplished
	· (Referral from CR) Coordinated relocation (from shelter in Denham Springs to LSU Medical Shelter) of 8 elderly, disabled, or homeless SN applicants. Shelter manager was “thrilled” with FEMA’s response to the request for assistance with these 8 evacuees. SNT will monitor progress.

	
	C.
	Staffing Needs:

   
	

	
	
	
	(3) Pending completion of DAE hiring process. 


	On site
	To be deployed 
	Needed

	
	
	
	
	9
	(3)
	0

	2.
	CURRENT ISSUES OR PROBLEMS:  None to report. 



	Note:
	(Referral from ERO) Began coordinating resources with VOLAG for food, clothing, medicine needed for two shelters with total of 70 evacuees, of which 27are mentally challenged. Will also request CR complete site visit for registering and  overall assessment of needs then follow-up.                                                                                           


	Section Reporting: 
	Inspection Services Coordinator

	Group Leader Contact Info:
	Barbara Copeland

	1.
	SITUATION:

	
	A.
	General Status of Operations
	Inspections are on going. There are 1258 housing inspectors in the field. 

	
	B.
	Major Priorities Accomplished
	

	
	C.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	0
	0
	0

	2.
	CURRENT ISSUES OR PROBLEMS: None.


	Section Reporting: 

	Sweep Registration Teams

	Group Leader and Contact Info:
	Jeanne Gallagher

	1.
	SITUATION:

	
	A.
	General Status of Operations
	· 9 SWEEP TEAMS ARE OUT ON THE FIELD TODAY

· The contractor has 65 people in the field

· We are supporting the Folgers effort with contract staff

· Training will be conducted this Saturday for all Fluor sweep team employees

· We are determining which Fluor staff is willing to travel to work in DRC’s

· We are hoping to place the remaining Fluor staff in JFO positions.

	
	B.
	Major Priorities Accomplished
	· We have completed all of the major shelters for registration intake

· We are working on registering applicants in special needs shelters

· We are finding most people are registered and have program questions

· The staff will be finished with registration intake on Saturday, September 24th
· Working with DFTO to set up training.  They only have a brief two hour HS overview.  Will need to coordinate with the NPSC for more extensive training

· Need to find assignments for the remaining Fluor staff once registration intake is completed this weekend.     

	
	C.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	109contractors / 3FEMA Staff
	0
	0

	2.
	CURRENT ISSUES OR PROBLEMS: None. 


	Section Reporting:
	NPSC Liaison

	Group Leader Contact Info:
	Pam Glasschroeder 703/785-2661

	1.
	SITUATION:

	
	A.
	General Status of Operations
	· Reviewing and resolving issues coming in from the DRCs and PPI group.

· Reviewing reports identifying problems / trends

· Provide updates on DRC status to NPSC Coordination Team

· Review and resolve inquiries from the NPSC Coordination Team

· Review cases from ERO, Congressional liaisons.

	
	B.
	Major Priorities Accomplished
	 Priorities:

· Corp of Engineers has updated Blue Roof toll free phone number (1-888-ROOF-BLU to include information for Louisiana residence. This information has been provided to NPSC/DRC staff
· Working with NPSC to identify how to register children who have loss their parents due to the disaster.
· Working on setting up DR 1603 MH/TT mailbox for PPI request form from NPSC help line staff

	
	C.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	1
	0
	1

	2.
	CURRENT ISSUES OR PROBLEMS:  




	Section Reporting: 
	ESF-6 Mass Care

	Group Leader Contact Info:
	Douglas Erickson 225-334-2103

	1.
	SITUATION:

	
	A.
	General Status of Operations
	· ESF-6 Mass Care continues to support this operation

· Reviewing pending ARF’s and checking on receipt of commodities

	
	B.
	Major Priorities Accomplished
	· Provided the operation with GIS friendly data in regards to availability and location of beds in all shelters north of the I-10/12 corridor as well as a list of shelters below the I-10/12 corridor.

· Long range (30-60 days) feeding plan.  Red Cross/ Salvation Army remain “status quo” and no additional federal resource needs at this time other than what is in place at this time.  Will reevaluate pending HURRICANE RITA impacts.

	
	C.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	3
	0
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 
· Louisiana state DSS  shelter lists can be obtained from ladonatedgoods@lhseop.louisiana.gov 

· ARC kitchens in the two Algiers locations have been asked to leave at the request of the Mayor.

· ESF-6 is trying to keep a handle on all shelter movement within the state.


	Section Reporting: 
	ESF-11

	Group Leader Contact Info:
	Georgina Castillo

	1.
	SITUATION:

	
	A.
	General Status of Operations
	USDA continues delivering food and facilitating delivery of nutrition assistance programs in affected States (including host States). 

USDA’s Food Nutrition Service (FNS) is delivering food and facilitating Food Stamp Program delivery in affected states (including host states).

Some activities include the following:

Food Stamp Program

Chart represents Food Stamp Program data related to disaster response provided to date; information is not all inclusive and figures are expected to change.

State

No. of Households

$ Value of Benefits

Arkansas

 7,147

$     2,024,693

Louisiana

         303,624

     $ 123,918,387
New Mexico

226

     $          41,824

Oklahoma

1,079

     $        273,886

Texas

124,988

$   41,299,747

WIC Program 

· A Mission Assignment for Louisiana WIC (Department of Health and Hospitals, Office of Public Health) was approved to deploy 31 volunteer WIC health professionals to assist in clinic operations.  They will be dispersed to metropolitan areas throughout the State where large numbers of evacuees have settled. 



	
	B.
	Major Priorities Accomplished
	Since the last reporting period, three trucks of mixed food items were sent to the American Red Cross.  


	
	C.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	3
	1
	0

	
	


	Section Reporting: 
	DRC OPS


	Group Leader and Contact Info:
	Don Baggett

	1.
	SITUATION:

	
	A.
	General Status of Operations
	DRC at Belle Chasse will be open daily from 7am – 5pm.  All other DRC’s will be open daily from 9am – 7pm.

 Fixed DRC sites now in 16 locations:

Caddo(Shreveport)

Ouacita(Monroe)

Rapides(Alexandria)

Lafayette(Lafayette)

Iberville(Plaquemine)

Plaquemines (Belle Chase)

Webster (Minden)

E. Baton Rouge (Baton Rouge)

Washington(Bogalusa)

Avoyelles(Marksville)

Washington (Franklinton)

St Tammany(Slidell)

Orleans(Algiers)

St James (Vacherie)

LaFourche(Thibodeaux)

St Charles (Boutte)

In addition, supporting 5 of the strike teams with registration at MDRC locations.

We have also secured leases on properties at the following locations and will announce openings ASAP(In order of priority):

Tangipahoa (Amite)

LaFourche(Raceland)

St John (LaPlace)

Jefferson(Gretna)

Ascension(Gonzales)

Iberia(New Iberia)

Point Coupee (New Roads)

Livingston (Denham Springs)

#1 Priority: To secure inspected, approved, and leased sites at the following Parishes:

E. Baton Rouge (Baton Rouge 2nd location)

Calcasieu (Lake Charles)

St Tammany (Covington)

W. Baton Rouge(Port Allen)

Terrebonne (Houma)

Sabine (Many)

Lincoln(Ruston)

St Landry(Opelousas)

Richland(Rayville)

Red River(Coushatta)

Jefferson Davis(Jennings)

Franklin(Winnsboro)

Evangeline(Ville Platte)

Cameron(Johnson Bayou)

Acadia(Rayne)

LaSalle(Jena)

Caldwell(Columbia)

Tensas(St Joseph)

Beauregard(DeRidder)

Morehouse(Bastrop)

Concordia(Vidalia)

Madison(Talullah)

St Martin(Breaux Bridge)

Jackson(Hodge)

Grant(Pollack)

Natchitoches(Natchitoches)

Vermillion(Abbeville)

Vernon(Leesville)

St Helena(Greensburg)

E Feliciana(Clinton)

W Carrol(Oak Grove)

Assumption(Napoleanville)

Winn(Winnfield)

More locations to follow.

	
	B.
	Major Priorities Accomplished
	

	
	C.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in

 
	# of Staff still needed

	
	
	
	
	422
	30
	200

	2.
	CURRENT ISSUES OR PROBLEMS:   

 All DRC Openings are currently suspended until we are assured that Hurricane Rita is no longer a threat to the State of Louisiana.  If the projected storm track holds, tentative plans are to open four additional DRC’s on Saturday, 9/24/05 and another four on Monday, 9/26/05.




	Section Reporting: 
	Housing

	Group Leader Contact Info:
	Juan Gil/Maggie D.

	1.
	SITUATION:

	
	A.
	General Status of Operations
	It is expected that some units in the Special Needs mobile home park in Pearl River should be ready for occupancy within the next three weeks.  

A representative from the office of the Mayor of Baton Rouge met with IA Housing staff today.  The Mayor’s office has begun identifying families who will be housed in units at the Groom Road Travel Trailer Group Site in Baker, LA.  The intent is to house families currently located in shelters in the East Baton Rouge area.

In anticipation of a potential evacuation due to Hurricane Rita, contact was made with Emergency Managers in the Parishes where applicants have been placed in FEMA travel trailers in commercial parks.  Emergency Managers confirmed that the parks were included in their evacuation plans.
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